

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Alhola, Sara; Chimbutane, Feliciano; Reinikka, Ritva

Working Paper

What shapes parents' attitudes towards bilingual education? Evidence from Mozambique

WIDER Working Paper, No. 40/25

Provided in Cooperation with:

United Nations University (UNU), World Institute for Development Economics Research (WIDER)

Suggested Citation: Alhola, Sara; Chimbutane, Feliciano; Reinikka, Ritva (2025): What shapes parents' attitudes towards bilingual education? Evidence from Mozambique, WIDER Working Paper, No. 40/25, ISBN 978-92-9256-599-2, The United Nations University World Institute for Development Economics Research (UNU-WIDER), Helsinki, https://doi.org/10.35188/UNU-WIDER/2025/599-8

This Version is available at: https://hdl.handle.net/10419/322165

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



What shapes parents' attitudes towards bilingual education?

Evidence from Mozambique

Sara Alhola,¹ Feliciano Chimbutane,² and Ritva Reinikka^{1,*}

May 2025

WIDER WORKING PAPER

wider.unu.edu

Abstract: This qualitative study explores parental attitudes towards bilingual education in Mozambique. To improve foundational learning, several countries in sub-Saharan Africa have re-evaluated their language-of-education policies and experimented with bilingual education. Community buy-in is essential for a successful bilingual programme, particularly in contexts where former colonial languages retain socioeconomic prestige. Despite variation across schools, our findings, based on focus group discussions in Maputo and Nampula provinces, show that a majority of parents support bilingual education. Initial scepticism often shifted when observing their children learning to read. However, parents consistently expressed concerns about their children not learning adequate Portuguese, the most common reason for opposing bilingual education. Furthermore, parents lacked information about the rationale, practices, and benefits of bilingual education—for which a remedy would be the school providing full and regular information. Apart from parental buy-in (demand side), for a successful bilingual programme effective implementation (supply side) is indispensable to produce robust learning results—the most critical factor shaping parental attitudes.

Key words: bilingual education, language of instruction, parental attitudes, education policy, implementation, focus groups, Mozambique

JEL classification: I21, I29

Acknowledgements: We thank Silvestre Cumbane, Domingos Machalele, Ânia Rossina da Silva (Maputo province), Francisco Victor Gaita, Laurinda Martinho, and Aurélio Luís (Nampula province) for excellent research assistance; Elias Zavale for technical assistance during the fieldwork; Mariamo Assane, Maurício Bernardo, Ezra Chambal Nhampoca, and Samuel Zucula for translation of the questionnaires and focus group transcripts; and Gervásio Chambo, Cláudia Ferreira da Costa, Edward (Sam) Jones, Manuel Lobo, Andre Loureiro, Ana Ruth Menezes, Lúcia Jose Nhampossa, David Noyes, Marcelyn Oostendorp, Anna Pörsti, Sirpa Sinervä, Laura Torvinen, and an anonymous reviewer for valuable comments and suggestions. Financial support from the Foreign Ministry of Finland is gratefully acknowledged. We thank the Embassy of Finland in Maputo for assistance during the fieldwork. We express our gratitude to the Mozambican parents, quardians, school directors, and deputy directors who contributed to this research.

This study is published within the UNU-WIDER project Academic excellence.

Copyright © The Authors 2025

UNU-WIDER employs a fair use policy for reasonable reproduction of UNU-WIDER copyrighted content—such as the reproduction of a table or a figure, and/or text not exceeding 400 words—with due acknowledgement of the original source, without requiring explicit permission from the copyright holder.

Information and requests: publications@wider.unu.edu

ISSN 1798-7237 ISBN 978-92-9256-599-8

https://doi.org/10.35188/UNU-WIDER/2025/599-8

United Nations University World Institute for Development Economics Research provides economic analysis and policy advice with the aim of promoting sustainable and equitable development. The Institute began operations in 1985 in Helsinki, Finland, as the first research and training centre of the United Nations University. Today it is a unique blend of think tank, research institute, and UN agency—providing a range of services from policy advice to governments as well as freely available original research.

The Institute is funded through income from an endowment fund with additional contributions to its work programme from Finland and Sweden, as well as earmarked contributions for specific projects from a variety of donors.

Katajanokanlaituri 6 B, 00160 Helsinki, Finland

The views expressed in this paper are those of the author(s), and do not necessarily reflect the views of the Institute or the United Nations University, nor the programme/project donors.

¹ Department of Economics, Aalto University School of Business, Finland; ² Department of Linguistics and Literature, Eduardo Mondlane University, Mozambique; * corresponding author: R. Reinikka, ritva.reinikka@aalto.fi.

1 Introduction

Despite the impressive expansion in enrolment in recent decades, many school children around the world spend years in school without significantly improving their literacy and numeracy skills. This learning crisis is estimated to affect about half of the learners in low- and middle-income countries. (UNESCO 2017; World Bank 2017).

Basic literacy and numeracy are foundational skills that all children should be able to acquire in early grades. Foundational skills are indispensable for any further learning as well as success in the workforce. Foundational learning is also more than rote learning—it requires conceptual understanding for problem-solving (RISE Research and Communications 2023).

This paper focuses on Mozambique, a low-income developing country that faces an enormous task in ensuring universal mastery of foundational skills in early grades. In a national learning assessment in 2016, only 4.9 per cent of Grade 3 students achieved the reading and writing skills as required in the curriculum (INDE/MINEDH 2017), a decline from an already very low achievement rate of 6.3 per cent in 2013 (INDE/MINEDH 2014). Another source of nationally representative evidence on foundational learning is the Service Delivery Indicators survey, which showed that, in 2018, less than half of Grade 4 students were able to recognize simple words, and less than 20 per cent were able to read a paragraph in Portuguese (Bassi et al. 2019).

Many reasons have been suggested to explain the poor learning outcomes in Mozambique. These include, among others, an exceedingly rapid expansion of the education system, high levels of student and teacher absenteeism, inadequate pre-service and in-service training opportunities for teachers, dominance of monolingual (Portuguese-only) basic education, poor nutrition, hunger, and poverty (UNESCO 2023).

According to research evidence, teacher quality is one critical factor in learning. Out of observable characteristics, subject-specific teacher skills have most consistently had a positive impact on student achievement. Most of this research, however, originates in high-income countries. For example, in a sample of 31 OECD countries, Hanushek et al. (2019) found that the increase of 1 standard deviation (SD) in teachers' literacy and numeracy skills was associated with a significant increase in student performance of 0.10–0.15 SD.

Using data from the 2014 SDI survey, which tested both Mozambican Grade 4 students for learning and their classroom teachers for literacy (Portuguese) and numeracy skills, Holvio (2022) estimated the causal impact of teacher content knowledge on student learning. In Mozambique, unlike in OECD countries, teacher content knowledge had little or no impact on student learning, on average. Yet, there was considerable heterogeneity in the results. Specifically, raising teacher content knowledge by 1 SD improved student achievement by 0.14 SD for students whose home language was Portuguese, similar to results in the OECD countries (Hanushek et al. 2019). Portuguese-speaking students were able to benefit from teachers' higher content knowledge, whereas the average

⁻

¹ The latest national learning assessment was carried out in 2024 but results have not been published at the time of writing. Some stakeholders have recently expressed a concern about the tests used in the Mozambican national assessments, specifically how accurately these tests are able to measure foundational skills.

Mozambican primary student—who started school with little or no understanding of Portuguese—was not. These findings were an important impetus for this paper and our focus on bilingual education as a potential way to bridge the language gap in early grades and, by doing so, significantly improve foundational learning in Mozambique.

The language of instruction is an important policy choice in education for any country. This is especially true in sub-Saharan Africa where there is a high degree of linguistic diversity, and where most countries are multilingual with an ex-colonial language—English, French, or Portuguese—chosen as the official language. Large majorities speak non-dominant languages at home but must learn in the official language at school or use it in other formal domains. Many countries have, therefore, opted for a transitional bilingual education policy, that is, teaching first in the home language (L1) and later introducing the second (dominant, ex-colonial) language (L2). This approach—often started as a pilot scheme—is based on education theory that recommends that, for the best learning results, schooling should begin in the language the child knows best (Ouane and Glanz 2010, 2011; Walter and Benson 2012). It also argues that education is most effective when teachers and students speak (well) the same language. In this transitional model, L2 becomes the only language of instruction in upper grades, whereas L1 remains as a subject.

Since independence in 1975, as in the earlier colonial era, Portuguese has been the language of instruction at all levels of education in Mozambique. The purely monolingual orientation of the first post-independence years began to give way to some form of multilingualism introduced in the early 1990s, especially in the new constitution and the National Education System Law (Chimbutane 2011). Bilingual education was launched, on an experimental basis, in 2003, and has expanded considerably in recent years mostly with foreign aid funding. However, bilingual education is yet to be offered universally in Mozambique.

Despite two decades of experimentation and expansion of bilingual education, Portuguese continues to enjoy the status of the language of socioeconomic mobility and sociocultural prestige in Mozambique (Chimbutane 2013; Chimbutane et al. 2022). These can be thought of as *ultimate* causes for attitudes towards bilingual education. But attitudes can also be shaped by *intermediate* causes, such as weak implementation (e.g., see Trudell and Piper 2014 on Kenya) or insufficient resources. Scarcity of trained bilingual teachers or lack of learning materials in African languages are often manifestations of such intermediate causes. There can also be little advocacy and engagement to obtain parental and communitity buy-in and participation in the design and implementation of education based on African languages (Stroud 2001; Chiatoh 2011 for Cameroon).

Poorly implemented and under-resourced bilingual education programmes, which have little or no community and parental buy-in or participation, do not offer an environment where students learn foundational skills effectively (Ouane and Glanz 2011). Such a weak learning environment may explain, at least partly, why parents² have often preferred to opt out from bilingual education in Mozambique (Chimbutane 2011) and elsewhere in sub-Saharan Africa (e.g., Bunyi 2008 for Kenya;

2

-

² In the Mozambican context, many children have guardians, often grandparents or other relatives, who interact with the school as their parents may be away as migrant workers, or deceased. In this paper, we use 'parents' to refer to both parents and guardians.

Rubagumya 2003 for Tanzania). This can create a negative feedback loop and prevent mainstreaming of bilingual education into the country's education system.

This study focuses on community and parental buy-in and participation in bilingual education at the school level in Mozambique. Specifically, using semi-structured focus group discussions, we provide an exploratory investigation into parents' attitudes towards bilingual education and factors that shape their attitudes. We pose the following research questions based on a set of assumptions:

- (i) Assumption: Parents' attitudes are an important factor in mainstreaming of bilingual education, its successful implementation, and, consequently, for improving foundational learning.
 - *Question*: Are Mozambican parents' attitudes towards bilingual education positive, negative, or indifferent? Have these attitudes changed over time?
- (ii) Assumption: Available information influences attitudes. The more information parents have on (the benefits of) bilingual education, the more positive are their attitudes. Question: How much information do parents have on bilingual education? What do parents expect from their children's school in terms of information and community mobilization?
- (iii) Assumption: Children's learning results and labour market prospects influence parents' attitudes towards bilingual education.

 Question: How do parents see the role of bilingual education in terms of their children's
 - learning results and labour market prospects?
- (iv) Assumption: Effectiveness of implementation can affect attitudes towards bilingual education. The better the implementation of the bilingual programme, the more positive parents' attitudes.
 - Question: How do parents perceive implementation of bilingual education at the school level, especially, regarding learning materials and teaching? What can parents' observations tell us about the effectiveness of implementation of bilingual education in Mozambique?
- (v) Assumption: General attitudes or sociocultural factors towards local languages affect parents' attitudes.
 - Question: What are parents' general attitudes towards local languages and how are these reflected in their attitudes towards bilingual education?

This study complements an earlier study that investigated bilingual education at the school level in Mozambique, specifically regarding the roles, attitudes, and actions of school directors, teachers, and local education officials—the 'supply side' of bilingual education (Chimbutane and Reinikka 2023). In contrast, this paper explores the 'demand side,' again, at the school level.

The rest of the paper is organized as follows. Section 2 summarizes recent literature on the (quantitative) impact of bilingual education on foundational learning in sub-Saharan Africa, including Mozambique, as well as findings on attitudes towards bilingual education across the world. Section 3 focuses on the Mozambican institutional context, including the sociolinguistic context, bilingual education policy and implementation, and community mobilization. Section 4 discusses the method, fieldwork, and data of the study. Section 5 presents our results. They cover parents' attitudes towards bilingual education and factors influencing these attitudes. The latter include information parents possess on bilingual education and views of the role of the school in informing parents; children's learning results and labour market prospects; implementation of bilingual education, especially

learning materials and teaching; as well as sociocultural aspects. Section 6 discusses the findings, while Section 7 summarizes and concludes.

2 Literature review

In this section we review, first, quantitative empirical evidence on the impact of bilingual education on foundational learning in sub-Saharan Africa³ and, second, literature on attitudes towards bilingual education across the world. Both reviews include available studies from Mozambique.

2.1 Impact of bilingual education on foundational learning

What is the impact of bilingual education on foundational learning? Can the use of children's home language in school make a difference for acquisition of literacy and numeracy skills? In this section we review several quantitative impact evaluations from sub-Saharan African countries that allow causal inference. Although somewhat less robust in terms of identification strategies, a few impact assessments are also available for Mozambique.

In neighbouring South Africa, Mohohlwane et al. (2024) present evidence on cross-linguistic transfer⁴ between home language (L1) and second language (L2), using results of two randomized evaluations of structured pedagogy programmes. These large-scale programmes targeted early grade literacy by a package of measures to improve classroom teaching. Both programmes had the same design, implementing organization, and duration. Both programmes were effectively implemented. The key difference was that one programme targeted the teaching of reading in L1, whereas the other targeted L2, which was English.

Both interventions had a positive effect on the language the programme targeted. The L1 intervention improved reading proficiency in L1 by 0.25 SD, whereas the L2 intervention increased reading proficiency in L2 by 0.11 SD. Interestingly, there were also impacts on the other language that was not targeted by the intervention. The L1 programme was found to improve reading proficiency in L2, too, but the L2 programme affected reading in L1 negatively (although only for lower-performing students). Based on these results, it seems cost-effective to prioritize learning to read in L1, even if becoming proficient in L2 is also an important policy objective.⁵

³ Many more countries in sub-Saharan Africa have adopted bilingual/multilingual education programmes than discussed here; we only include the programmes evaluated—quantitatively and causally—for impact.

⁴ Cross-linguistic transfer embodies learners' use of linguistic knowledge of their first language to leverage the learning of a second language.

⁵ A small caveat is that these results are derived from two experiments in two different populations. Although the two groups of students were similar in terms of socioeconomic status and education outcomes, their home language was different and, therefore, could potentially have had somewhat different responses to the treatment.

In Cameroon, Laitin et al. (2019) evaluated an experimental L1 programme for impact on foundational learning.⁶ This was also a transitional programme implemented by an international non-profit organization. The evaluation found that, compared with the control group, participating students exhibited large gains in both maths and the second language (L2), which was English and taught as a subject in early grades. The gains amounted to 1.17–1.71 SD (maths) and 0.68–1.53 SD (English). The authors explain these (unusually) large impacts by primary teachers' poor English skills, which meant that learning in monolingual (English-only) classes was 'shockingly low'. In Grade 5, however, when all students attended monolingual classes, any advantage bilingual students had gained in early grades seemed to have dissipated, and students' test scores for both the treated and control group were low.

In Guinea-Bissau and Niger, Hovens (2002) compares test scores in foundational skills of experimental bilingual schools with those of monolingual schools. In Guinea-Bissau the bilingual pilot programme used an abrupt transition from local language to Portuguese in Grade 3, whereas in Niger the transition to French was gradual. In both countries, resources were similar in both types of schools; only parents' socioeconomic status was lower in bilingual schools, which could act as a disadvantage in learning. The study found little difference in learning between bilingual and monolingual schools in Guinea-Bissau. In Niger, pupils who started in L1 could read and write better both in L1 and L2. In both countries, rural children and girls gained most from bilingual education. The country comparison also suggested that the earlier the bilingual programme starts teaching L2, and the later it introduces L2 as the medium of instruction, the better the results.

Ethiopia offers an example of late-exit bilingual programme in which English was introduced as the medium of instruction after Grade 8 and, in some cases, after Grade 6. Heugh et al. (2007) compare the scores of Grade 8 students in the national assessment in regions which used local language as the medium of instruction to those who were taught in English.⁸ They show that students in mother tongue education performed better in mathematics and sciences than students in English-medium education. Furthermore, Heugh (2009) shows that students with eight years of mother tongue education had higher overall scores across the curriculum than students with six years, although the latter performed better in English. These results underscore the advantages of an extended use of L1 as medium of instruction compared with submersion (in which use of L1 is prohibited), L2-based monolingual models, and early-exit transition models of bilingual education.

In Mozambique, the earliest experiment of bilingual education⁹ was implemented in the 1990s. A qualitative evaluation during its final two years (Grades 4 and 5) demonstrated that students had benefited greatly in terms of classroom participation, self-confidence, and biliteracy (Benson 2000).

⁶ The allocation of control and treatment status was not assigned randomly but, instead, through a heuristic matching procedure. It produced treated and control schools and students which were balanced along the available set of characteristics.

⁷ In Guinea-Bissau a total of 1,214 early grade students were tested in 32 schools, while in Niger 1,664 students were tested in 36 schools. Students were selected randomly and independently from their teachers. The tests were implemented both in L1 and in L2.

⁸ The study compares student scores in the upper cycle of primary education. All schools had mother tongue as the medium of instruction in the first cycle of primary education (Grades 1–4).

⁹ *Projecto de Escolarização Bilingue em Moçambique* was implemented in Gaza (Changana-Portuguese) and Tete (Nyanja-Portuguese) provinces in 1993–97.

Although the quantitative evaluation of this experiment faced design issues—non-random selection of participants, abrupt transition from L1 to L2, and implementation hick-ups—it still provided enough useful evidence encouraging government to launch its first bilingual programme in 2003. When tested for learning, the majority of bilingual students earned passing scores in L1. In the written Portuguese test in Grade 4, bilingual students were clearly behind monolingual students, although top performers in both groups had similar scores. However, one additional year made a difference for bilingual students. When tested again in Grade 5, in one province bilingual students averaged slightly higher than their monolingual peers. In the other province, while remaining behind, bilingual students cut the prior year's difference in half.

Ngunga et al. (2010) assessed the learning achievement of Grade 3 bilingual and monolingual students in Mozambique.¹⁰ This (small) study included the assessment of oral, reading, writing, and numeracy skills. Overall, bilingual students achieved better results in reading, writing, and mathematics (tested in L1) than their peers in monolingual education (tested in L2). In contrast, monolingual students demonstrated better oral skills in Portuguese.

To conclude, impact assessments confirm that bilingual programmes, typically experimental and externally funded, have delivered significant gains for early learning in sub-Saharan Africa. The Cameroon study—where the unusually large positive effects dissipated in upper grades—gave us a hint of the realities that experiments can face when mainstreamed. Therefore, one must ask: How do experiments scale-up as part of the country's education system, with domestic finance? What about parents' attitudes towards expansion of bilingual education beyond experiments? Do they support or resist it?

Similar questions are also posed in a recent review paper of several francophone West African countries (Ball et al. 2024). Importantly, the review argues that successes (or failures) in scale-up depend on the context. What works in one country or community may not work in another. To realize the benefits of bilingual education, its scale-up faces a broad range of social, cultural, and political factors, and institutional capacity issues specific to the country or community. All these factors influence programme design, implementation, and parents' attitudes.

2.2 Attitudes towards bilingual education

Bilingual education has produced mixed reactions from parents and communities worldwide, ranging from high demand to complete rejection. In cases where a language is considered to be of lower social standing than another (dominant) language, high demand for bilingual education tends to be associated with *integrative motivation*, such as acquiring cultural knowledge or wanting to belong to a group. Low demand for bilingual education using a low-status language tends to be associated with (lack of) *instrumental motivation*, such as (expected) poorer performance in examinations, limited opportunities for career advancement, or lower earnings (e.g., Chimbutane et al. 2022; Chonane 2024; Moses 2000; Terra 2021).

6

.

¹⁰ The programme was implemented in Gaza province with the support of UDEBA-Lab, a non-governmental organization, in 2003–09.

Limited support for bilingual education has been observed in many multilingual contexts. These include the United States (Moses 2000), India (Sridhar 1994), Malaysia, and Singapore (Gupta 1997). In sub-Saharan Africa, studies from Kenya (Bunyi 2008; Trudell and Piper 2014; Manyonyi et al. 2016), South Africa (Banda 2000; Phindane 2015), and Tanzania (Rubagumya 2003) show similar results. Parents felt that the use of a low-status language in education delayed or hampered their children's integration into the dominant language and culture, which was perceived as a prerequisite for socioeconomic mobility. In other words, according to research evidence, *instrumental motivation* seems to have dominated attitudes. In extreme cases, parents equated bilingual education to inferior education or saw it as a way to deny their children access to the language of privilege (Bamgbose 1999; Gupta 1997). Instrumental motivation may help explain why many parents in post-colonial contexts, such as Mozambique, prefer to enrol their children in public or private schools that teach in European languages. It is considered the fastest and most effective way to assimilate into the dominant language and culture. Especially urban and middle-class parents demonstrate this preference (Chimbutane and Gonçalves 2023; Sumich 2018).

However, there are also studies that show how the use of African languages in education helped bridge the gap between rural schools and surrounding communities. For example, following the introduction of first bilingual classes in Mozambique, Kitoko-Nsiku (2007) showed how bringing together teachers, students, and parents, and making the school a participatory centre for learning, affected attitudes about the entire educational environment positively. The use of local language enabled classroom interaction and facilitated the mobilization of community knowledge. Chimbutane (2015) showed similar findings nearly a decade later.

In Mozambique, a few studies have asked whether sociocultural gains 11 are enough to sustain bilingual education in the post-colonial context. Chimbutane (2013) found that, to gain status and flourish, bilingual education must be effective and successful not only in local languages but also in Portuguese, which is seen as a gateway to employment opportunities, economic advancement, and power. In other words, bilingual education can sustainably succeed if it is able to deliver not only sociocultural but also academic benefits that lead to socioeconomic gains (e.g., Chimbutane 2011, 2013; Mataruca 2014). Uneducated parents—who have not gone to school—were found to be more supportive of bilingual education, whereas educated parents supported monolingual education, emphasizing the importance of learning Portuguese (Henriksen 2014).

3 Institutional context

In this section we discuss, first, the sociolinguistic context of Mozambique; second, the government's bilingual education policy and its implementation, and third, experience in community mobilization to support bilingual education.

¹¹ In terms of sociocultural gains, bilingual education can help legitimize marginalized languages and cultures, maintain and develop local languages, and integrate local knowledge and communities in schools (Chimbutane 2011).

3.1 Sociolinguistic context

Mozambique is a multilingual polity, with over 20 Bantu languages ¹² spoken across the country, in addition to Portuguese and a few foreign languages. According to the 2017 national population census, 82 per cent of the population reported speaking a Bantu language and 58 per cent Portuguese (Instituto Nacional de Estatística 2019). About 80 per cent of the citizens speak a Bantu language as their mother tongue in contrast to 16.6 per cent whose mother tongue is Portuguese.

Despite this sociolinguistic profile, Portuguese, a former colonial language, maintains its privileged status as the sole official language of the country. In contrast, no official status has been granted to Bantu languages that continue to be confined to informal domains and functions. The argument for the choice of Portuguese as the official language at independence in 1975 was to ensure national harmony, modernization, and internationalization. The unifying ideology underpinning this choice was epitomized by the declaration of Portuguese as *língua da unidade nacional* (i.e. the language of national unity).

Consistent with this choice, Portuguese is the de facto language of formal education, socioeconomic mobility, and social distinction, notwithstanding some changes taking place since the 1990s (see Section 1). This may explain, by and large, why many parents, in particular urban and middle-class parents, prefer to socialize and formally educate their children exclusively in Portuguese, even if they continue to be socioculturally attached to their ancestral languages and cultures (Chimbutane and Gonçalves 2023).

3.2 Bilingual education policy and implementation

Bilingual education in Mozambican primary schools was introduced two decades ago on an experimental basis. Since then, the programme has expanded gradually, from 14 schools in 2003 to 3,225 schools in 2021 (UNESCO 2023). Currently, almost a quarter of all 13,800 primary schools offer bilingual classes (MINEDH 2022). Despite the substantial expansion—particularly in the two most populous provinces, Nampula and Zambézia, thanks to a series of projects supported by the United States Agency for International Development (USAID)—country-wide coverage remains uneven.

The Mozambican government's *Bilingual Education Expansion Strategy for 2020–29* lays out the framework for a gradual expansion of bilingual programme throughout the country (MINEDH 2019). Today bilingual education is offered in 19 local languages. The strategy calls for expansion to linguistically homogeneous communities, identification and training of teachers, community mobilization to ensure understanding and buy-in from parents and community leaders, and training of school directors and district education officers for effective management of the bilingual approach.

¹² These languages are also referred to as 'Mozambican languages' or 'local languages'.

The primary school curriculum was also reformed recently to focus more on foundational skills (MINEDH 2020).¹³ This is in recognition that many learners leave primary school without basic literacy and numeracy. Implementation of the revised curriculum, which started in 2017 from Grade 1, moved ahead one grade per year. The revised Grade 6—the final year of primary school today—curriculum was introduced in 2022.

Mozambique uses the transitional bilingual education model that starts in the early grades with home language (L1) as the medium of instruction along with lessons in Portuguese as the second language (L2). In Grades 1 and 2, as shown in Table 1, students learn reading, writing, and maths in local languages but Portuguese only orally. In Grade 3, students begin to read and write in Portuguese. With each grade, instruction time in Portuguese increases, with local languages maintained as a subject.

Table 1: L1 to L2 transition in bilingual education, Grades 1-6

Discipline	Grade 1	Grade 2	Grade 3	Grade 4	Grade 5	Grade 6
Portuguese language	L2 oral	L2 oral	L2 oral + L2 reading and writing	L2 reading and writing	L2 reading and writing*	L2 reading and writing*
Mozambican languages	L1	L1	L1	L1	L1	L1
Mathematics	L1	L1	L1	L2*	L2*	L2*
Social sciences	_	_	_	L1	L1	L2*
Natural sciences	_	_	_	L1	L2*	L2*
Visual education, arts and crafts	_	_	_	_	L1	L2*
Physical education	L1	L1	L2	L2	L2	L2
% L1	80	80	67.7	43.3	16.7	6.7
% L2	20	20	30.3	56.7	83.3	93.3

Note: * indicates that a similar teaching programme and books as for monolingual students are used.

Source: authors' compilation based on MINEDH (2020).

Most bilingual teachers in Mozambique have been trained to teach in the monolingual programme and have to adapt to teach in Mozambican languages. In an early study of teacher preparedness, Patel (2006) notes that teachers were unfamiliar with the theories of bilingual education and lacked the command of the curriculum. A decade later, bilingual teachers still lacked pedagogical knowledge and skills in bilingual education (Terra 2021) and continued to use monolingual pedagogies (Chimbutane 2015). This has also been a concern of Mozambican parents (Mataruca 2014).

3.3 Community mobilization

In the past, language-in-education policies used to be decreed in a top-down manner, rather than taking into account the needs and opinions of parents and communities. This approach began to

¹³ To reserve more time for foundational skills, English language and music education were dropped from the revised curriculum.

change globally around the time when Mozambique first introduced its pilot in 2003. For bilingual programmes to be successful, Stroud (2002) suggested that community ownership, along with local input and community management, should be maximized. Bilingual education policies in Mozambique have reflected this recommendation, and various approaches have been tried out by government, civil society organizations, and development partners engaged in bilingual programmes. These include awareness campaigns, use of media such as community radios, dissemination of learning results, and involvement of local administrators and leaders (Chonane 2024; Henriksen 2014; Kitoko-Nsiku 2007; MINEDH 2020).

A key objective in these various approaches has been to change negative attitudes towards bilingual education, especially by informing communities and parents of the advantages of instruction in both African languages and Portuguese. However, several studies suggest that community mobilization efforts may not have produced a significant change in people's attitudes (Chimbutane 2011, 2013; Chonane 2024; Cumbane 2020). These studies suggest that, to achieve its goals, community mobilization efforts should be 'more dynamic and effective' than they have been so far.

For example, Chimbutane (2011, 2013) found that, when mobilization is effectively implemented, local communities in Mozambique tend to embrace bilingual education, which is primarily seen as a way to promote, preserve, and develop African languages and associated cultures. In Maputo province, Cumbane (2020) found that community participation was negligible as communities remained at the margins of teaching and learning processes; bilingual education did not contribute much towards their empowerment. Yet, community participation was significant in other areas, such as school construction and maintenance of school infrastructure.

In a recent study, also in Maputo province, Chonane (2024) found a weak effort in community mobilization. Most parents were not aware of the purpose of bilingual education, or how it was implemented in their children's school. Parents reported that it was only at the opening ceremony of the school year when parents had learnt about their children's assignment to a bilingual class.

In sum, based on available evidence, community mobilization remains weak in Mozambican bilingual education.

4 Method, fieldwork, and data

As this study is exploratory in nature, we attempt to answer our research questions (Section 1) using focus group discussions. In other words, we try to understand what, broadly speaking, shapes or underlies parents' attitudes towards bilingual education in Mozambique. This also helps lay the groundwork for subsequent quantitative analysis.

Researchers using focus group discussions as their (qualitative) research method typically prepare in advance a set of targeted questions designed to elicit collective views from participants about the topic under study. Interaction is fundamentally important during a focus group discussion. Facilitators help generate in-depth conversations, enabling, to the maximum extent, everyone's participation within the group. They encourage participants to express themselves in their own language. Participants are discussing together rather than reacting to interview questions individually. Focus group data are, therefore, a sum of the structured protocol of targeted questions, presentation of the

facilitators, and the interaction with and between participants (e.g., see Parker and Tritter 2006; van Ingelgom 2020).

The purpose of focus group research is not to infer but to understand, not to generalize but determine a range, and not to make statements about the population but to provide insights about how people in the groups perceive a situation (Krueger and Casey 2009, quoted in van Ingelgom 2020).

Participants are invited to engage in focus groups because they have something in common with each other—in our case parents of Mozambican primary school students. In a successful focus group session, a kind of momentum is generated that allows underlying opinions, meanings, feelings, attitudes, and beliefs to emerge alongside individual experiences (Parker and Tritter 2006). At the collective level, what often emerges from a focus group discussion is a number of positions or views that capture the majority of the participants' standpoints. Focus group discussions seldom generate consensus but create a number of views that different proportions of the group support.

The fieldwork for this study was carried out during September and October in 2023 in Maputo and Nampula provinces. Each province had a (separate) team of experienced researchers who also spoke the local language. The teams facilitated focus groups using a predetermined set of themes or questions (Section 1). In both provinces, focus groups were organized in three districts, in which two schools per district were selected, resulting in a total of 12 school-based focus groups (Table 2). The school director together with teachers extended an open invitation to parents to participate in the focus group discussions that were held in the school premises.

In Maputo province—surrounding the capital city in the southernmost area of Mozambique—the selected schools were located in the Manhiça, Matutuíne, and Magude districts. Two of the schools used Ronga as the home language in bilingual education, whereas the remaining four schools used Changana. Nampula is the most populous province in the country, with Macua as the main local language. As mentioned earlier, USAID has supported bilingual education for the past decade in Nampula province. USAID's initial focus was on local language literacy. Subsequently, the focus shifted to supporting government's bilingual education programme as a whole. 14 The selected schools for our study were located in Mugovolas, Ribáuè, and Murrupula districts.

All 12 schools offered bilingual education, but not necessarily at every grade level (Table 2). The schools had also monolingual (Portuguese-only) classes. In the six schools in Maputo province, bilingual education had been introduced between 2017 and 2019. In Nampula province bilingual education was introduced more recently: two schools in 2019, three schools in 2020, and one school only in the beginning of 2023. Most schools had bilingual classes in Grades 1–4. In Nampula province, once the school introduced bilingual education, it was the only modality offered (i.e. no

11

thousands of other USAID projects across the world.

¹⁴ The USAID-supported Aprender a Ler! sought to improve students' reading performance through teacher training and coaching, development and use of complementary reading material, and improved school management and district supervision. The subsequent Vamos Ler! programme supported MINEDH in expansion of bilingual education to 2,010 schools in Nampula and Zambézia provinces (USAID 2022). The most recent project SABER had plans to expand to Niassa and Cabo Delgado provinces and to add 1,395 new schools to bilingual education. However, in February 2025 the US government abruptly terminated this project along with

parallel monolingual classes), whereas the schools in Maputo province continued to offer parallel monolingual classes.

Table 2: Characteristics of the sample schools

Characteristic	(Number of schools)
All schools	12
Region	
Maputo	6
Nampula	6
Bilingual classes offered	
Grade 1	12
Grade 2	9
Grade 3	8
Grade 4	9
Grade 5	7
Grade 6	5
Language of instruction in bilingual education	
Changana	4
Ronga	2
Macua	6
Year bilingual education was introduced	
2017	2
2018	3
2019	3
2020	3
2023	1

Source: authors' compilation based on study data.

Our target was to have a total of ten parents agree to participate in each focus group. However, the number of attendees ended up being larger in some schools as more parents than expected responded positively to the invitation. We felt that it was important to include all parents who had come to the school with the purpose of participating in the focus group. The only prerequisite to join the group was that the parent (or guardian; see footnote 2) had at least one child currently enrolled in the school, whether in a bilingual or monolingual class.

Table 3 shows descriptive statistics of the focus group participants. As can be seen, there were more participants in Nampula province as the focus groups were larger there. However, the total number of focus groups was the same—one per school—in both provinces. The majority of parents participating in the focus group discussion were female; the average age of participants was 35 years. More than three quarters of parents had a child in bilingual education and most of them also had a child in monolingual education.

Table 3: Characteristics of focus group participants

Characteristic	
Number of participants (total)	162
Region (%)	
Maputo	38.3
Nampula	61.7
Gender (%)	
Female	63.3
Male	36.7
Age (years)	
Average	35
Min	18
Max	68
Language (%)	
Changana	25.8
Ronga	11.3
Macua	62.9
Children's current language modality (%)	
Parents with children in bilingual education	77.8
Parents with children in monolingual education	56.8

Source: authors' compilation based on study data.

Facilitators informed focus group participants about the remit of the research, main topics for discussion, and processes of data transcription. They also reaffirmed confidentiality and anonymity. One member of the research team was assigned to take notes throughout the discussion. With the permission of participants, discussions were audio recorded so that the team could produce a full (verbatim) transcript from each focus group discussion. Afterwards, the team of facilitators held an internal debriefing that was summarized in a separate note. Verbatim transcripts were prepared in the local language and subsequently translated into Portuguese and English.

Our data analysis followed the general principles of qualitative data analysis. Specifically, we analysed the verbatim transcripts from the 12 focus group discussions systematically using the thematic analysis approach (e.g., van Ingelgom 2020). This included preparing topic matrices and summaries. With our research questions (Section 1) as the point of departure, we primarily adopted a deductive approach to structure the analysis. The predefined themes were used to identify and compare the views arising from the discussions. However, we also remained open to new insights by identifying and mapping frequently emerging themes. This allowed us to capture valuable parental perspectives that were not initially anticipated.

In addition, after having completed our deductive analysis, we used artificial intelligence (AI)¹⁵ to examine the transcripts to identify patterns and generate insights related to our research questions. A comparison between our deductive analysis and the AI results did not reveal any material differences. Thus, AI served as an additional verification tool, providing us with an extra layer of confidence in the findings.

-

¹⁵ Aalto Al Assistant, which is linked to the ChatGPT language model.

5 Results

This section reports our findings. The overarching assumption of this study was that parents' attitudes are an important factor for a successful bilingual education programme which, in turn, would help improve exceedingly weak foundational learning in Mozambique. First, we explored whether parents' attitudes towards bilingual education were positive (supportive), negative (not-in-favour), or indifferent, and whether these attitudes have changed over time.

Second, we explored a number of factors expected to influence parents' attitudes towards bilingual education, including information available to parents, student learning results, expectations about students' labour market prospects, quality of implementation of bilingual programmes, and parents' sociocultural attitudes towards local languages.

5.1 Parents' attitudes

We found that, overall, parents' attitudes towards bilingual education varied across schools. Roughly half of the focus groups were unequivocally supportive of bilingual education. Among the other half, some focus groups expressed qualified support, whereas in a few schools parents' attitudes were decidedly negative. This variation was more notable in Nampula, where there was strong opposition in some schools.

In the schools where we witnessed unanimous support, parents were happy that their children learnt to read and write in their home language. Some parents even suggested that bilingual education should be the principal modality throughout primary education. Others believed that their children would face fewer difficulties in life when they learnt both the local language and Portuguese.

I liked the fact that there are two languages. Because the first one is our doctrine at the country level and Portuguese is a foreign language. . . . When a child is born, he must begin with his doctrine, then move on to international languages, because all these things will make it easier for him. (Man, 52 years)

While still mostly positive, in a few other schools, parents had some reservations, and their support was more nuanced. The main concern was that students may not learn enough Portuguese.

I think it [bilingual education] is important for those who like it. But we also have some disapproval as we already speak Changana at home, and we go to school because we want to learn Portuguese. (Woman, 39 years)

As said, in a few schools scepticism was pronounced, as parents saw little or no benefit from bilingual education. Many would have preferred their children to learn exclusively in Portuguese. The concern about students not learning enough Portuguese was urgently felt by parents—much more so than in the 'partially supportive' focus groups.

I don't like Macua being taught to my young son because I sent him to school so he could study Portuguese. At home we have our language, which is Macua, our mother

tongue. So, it does not make sense that here at school they study Macua, too, without studying Portuguese. I don't like it! (Man, 49 years)

In light of our evidence, it is not easy to determine whether attitudes had changed over time. A similar mix of attitudes had been reported in studies carried out (in Gaza province) over a decade ago (Chimbutane 2011, 2013), indicating little change in attitudes. At the same time, many focus group participants reported that their initially critical attitude had shifted. As experience of bilingual education accumulated, many parents had come to accept it. This could indicate an improvement in attitudes over time. At the very least, it illustrates that parents' attitudes are not static.

5.2 Information

Regarding factors that shape attitudes, we started by exploring whether information that parents have on bilingual education—on its rationale, implementation details, transition from home language to Portuguese—influenced their attitudes towards it. Our assumption was that the more information made available, the more supportive the parents. We asked parents participating in focus groups what they knew about bilingual education currently as well as before enrolling their children in school or before the school implemented bilingual education.

In most focus groups, parents reported that they had not been part of the decision-making when the school had initially adopted bilingual education. The lack of engagement meant that parents were often not convinced about the benefits of bilingual education when it was adopted. We also asked parents who ultimately decided the placement of children either in monolingual or bilingual class. In nearly all schools, parents said that the decision rested solely with the school. In some schools, parents had been asked about their language preference or their child's mother tongue, but this had seldom translated into a class placement decision.

This was not something we were informed about. We just came and saw our children receiving books in Macua. (Man, 46 years)

We found that parents often possessed little information on bilingual education when they came to enrol their children in school; many parents said this had led to scepticism and misconceptions. For example, the transition from local language to Portuguese was often unclear—some parents had thought that teaching would be solely in the local language throughout primary school.

We thought that children learnt in Ronga until the end of primary school. We were concerned about the higher grades because, in the secondary school, they learn in Portuguese and would face difficulties. (Woman, 53 years)

The information parents had on bilingual education remained, overall, highly variable across schools, and often incomplete. In schools where they had received more information (relative to other schools)—as expected—parents were more supportive of bilingual education. They understood the pedagogical rationale relatively well. A good example was a school where a separate meeting had been arranged at the beginning of the year to explain both the pedagogical benefits and implementation practicalities. Participants appreciated the fact that the parents whose children attended monolingual (Portuguese-only) classes were also included in the meetings.

However, there were focus groups where participants stated that they had received no information whatsoever about bilingual education from the school, or any other source. Some parents even learnt the specifics of bilingual education for the first time during our focus group discussion! Others said that they had only become aware of it when their children brought home learning materials that were in the local language instead of Portuguese.

Many parents don't know why bilingual education was introduced. Others only hear that Macua is being learnt, but the reason why instruction is in two languages is not known. I saw my children coming home with their Macua books, but I didn't know what was intended with this programme. (Man, 24 years)

In these schools (with less information), parents seemed unaware of any pedagogical benefits. In one school, parents had been told that the motivation of bilingual education was to improve knowledge of the local language and, thus, to help children succeed in places that used the same local language. Parents in that school were strongly opposed to bilingual education.

Across focus groups parents maintained that prevailing misconceptions were due to a lack of information, not only among parents but the whole community. This hindered acceptance of bilingual education and reduced support to it.

We should inform the entire community and a meeting should be held to open the school year and invite everyone to participate to clarify this further. . . . It's for you [school] to clarify to us the advantages of children learning in two languages . . . some parents simply refuse because they do not know the advantages. (Man, 53 years)

My opinion was that when the time comes for the opening meeting of the school year, regardless of whether you have a child or not, you should be present so that the teachers can inform us for everyone to know. (Woman, 43 years)

Parents expressed concerns about potential stigmatization of children in bilingual programmes or their families. This was because of the perceived inferiority of local languages compared with Portuguese and the perceived slower pace of bilingual students learning Portuguese (see also Chimbutane 2021). Parents felt that addressing these concerns was crucial to ensuring both parents' and children's comfort and willingness of children to continue going to school.

Overall, majority of schools had made an effort to inform parents, often in a meeting held at the beginning of the school year, but the information provided lacked specifics, and the meeting was a one-off event.

5.3 Student learning

Across schools, parents indicated that their attitudes were influenced, most of all, by their children's progress in learning—progress that the parents themselves had witnessed. Many parents explained that, despite their initial scepticism, they had changed their mind and accepted bilingual education when they had noticed that their children were able to read in the local language soon after starting

school. Reading skills—rather than, say, numeracy—were what parents appreciated most. A number of participants also noted that their children had acquired reading skills earlier in a bilingual class compared with a monolingual one.

It is easier for our children to learn Changana and Portuguese, because those in Grade 1 are already reading. My son can already read, which really surprises me. (Woman, 29 years)

We see results. At first, we didn't accept it [bilingual education], because at home they already speak Changana. For them to study in *Changana* here at school, it wasn't good for us. But now, we can see that they are studying well, because even in church they can already read the Bible [in Changana]. (Woman, 45 years)

As in earlier studies (e.g., Chimbutane 2015), many participants pointed out how learning in the local language connected the school better with everyday life, and reduced the risk of dropping out before finishing primary school. In several schools, initial resistance and scepticism had softened when parents had observed, in addition to improved learning, their children's more active participation in school work.

In fact, learning to read has been the most important aspect of bilingual education for Mozambican parents ever since the very first experiment in the 1990s. Already then parents expressed great pride in students' accomplishments in the mother tongue (Benson 2000). Just like in our study for more than a quarter century later, parents cited examples of the usefulness of their children's literacy, which ranged from reading the Bible aloud during church services to writing letters to family members working in South African mines. Then and now, parents want their children to learn.

Parents who reported positive learning results in both local language and Portuguese were the most supportive of all of bilingual education.

My son studied Macua, but when he reached Grade 5, I suddenly heard him speak Portuguese (laughter) and I asked him: "How does that work?" He answered: "I already stopped studying Macua and entered in Portuguese language." I wished him strength; maybe he can help us out in the future. (Woman, 38 years)

However, doubts about Portuguese language acquisition created resistance towards bilingual education. In the focus groups where opposition was strong, participants argued that they had enrolled their children in school for them to learn Portuguese. Negative experiences—such as witnessing a child going through bilingual education without learning to read and write properly in either language—increased parents' pressure on the school to switch to monolingual education. Many parents reported that they had not seen sufficient Portuguese acquisition in bilingual classes, putting their children in an unfavourable position relative to those in monolingual classes. Children's proficiency in Portuguese was also considered necessary in social settings—some parents had felt

_

¹⁶ Only 48 per cent finish primary school in Mozambique today.

uncomfortable when their children could not speak Portuguese with relatives. Some parents were even convinced that their children were not taught in Portuguese at all.

The challenges we're experiencing at school is that we cannot hear the students being taught in Portuguese. This problem is really breaking our hearts. (Man, 31 years)

While parents may have accepted the pedagogical rationale—that the use of local language in early grades makes learning basic skills easier—they remained doubtful about its *overall* benefits. Again, this was largely because of the concerns that their children would not be able to learn sufficient Portuguese during their schooling and, therefore, not benefit from socioeconomic gains that come from mastering the dominant language. These concerns have appeared and reappeared in many studies since the introduction of bilingual education (e.g., Chimbutane 2011, 2021).

The varied parental experiences suggest that the learning benefits of bilingual education have not materialized in the same way in all schools. A possible explanation could be the timing of implementation across our sample of schools, allowing some schools more time to adapt and ensure pedagogical benefits. However, a closer look at the data revealed that the year of implementation did not correlate with parents' views on the learning benefits of bilingual education.

5.4 Labour market prospects

Regarding their children's prospects in the labour market, parents across schools emphasized, again, the importance of learning Portuguese. This included parents who were in favour of bilingual education and those who opposed it. Although participants in some focus groups expressed hopes for a bigger role for local language in the labour market, everyone highlighted the necessity of their children mastering Portuguese to succeed in the workplace. A common refrain among parents was: 'Portuguese is needed for getting a job', highlighting the importance of instrumental motivations and socioeconomic benefits. No other subject or foundational skill—say, numeracy—came up as a prerequisite for finding a job except for Portuguese.

The truth is that studying Ronga [is not as advantageous] as Portuguese. When looking for a job, if you speak Portuguese, you have an advantage over someone who speaks Ronga. Portuguese is at the top in relation to Ronga. (Woman, 55 years)

Some parents felt that their children were not learning Portuguese at all, and that with the knowledge of a local language only, their children would never be employed. Instead of bringing value to schooling, they considered bilingual education as a serious hindrance for their children's future. In a couple of focus groups, parents even speculated that decision-makers' lack of transparency regarding bilingual education stemmed from their awareness of its limited overall benefit, denying bilingual students job opportunities. Parents argued that, in their experience, government made decisions without community consultation and, therefore, it did not serve the best interests of the population. Bilingual education was seen as a manifestation of this situation.

Yet, parents who viewed bilingual education as a means for their children to learn both the local language and Portuguese highlighted its advantages for their children's future. Proficiency in two languages, they noted, would open more doors and opportunities.

Finally, apart from Portuguese, a number of parents in several focus groups highlighted the importance of studying English as well even if it was discontinued in the most recent revision of the primary curriculum. Parents felt that English was important for employability.

We still don't know, if there will be jobs for those learning Changana. The child can study in a bilingual class but, in the end, jobs require English. Institutions require English and Portuguese. Children should know English and Portuguese . . . but without losing their Changana. (Woman, 46 years)

5.5 Implementation

In our previous study, school directors and teachers argued that monolingual education was much better resourced than bilingual education, which lacked learning and teaching materials and trained teachers (Chimbutane and Reinikka 2023). A number of other studies have also shown that the bilingual programme in Mozambique has suffered from scarcity of teaching and learning materials (Manuel et al. 2024; Ngunga 2011; Terra 2021; USAID 2017). In this section, we explore how Mozambican parents perceived implementation and resourcing of bilingual education.

Instead of just resources in bilingual education, focus group discussions revealed a general lack of resources both in monolingual and bilingual classes. Discussions mostly concerned students' textbooks, as parents had little or no information on teachers' resources. While in some schools parents reported their children having enough textbooks in bilingual classes, especially in early grades, in other schools parents complained about shortages. In schools where bilingual education had been introduced recently, parents reported a lack of trained teachers and learning materials in the local language.

We have a very big lack of teachers and materials here. I think that the programme that government initiated, in order not to fail, should bring teachers and materials for all the students and train the teachers. (Man, 43 years)

Across focus groups there was a widespread complaint about lack of Portuguese textbooks and their poor condition in both monolingual and bilingual classes. Some parents, particularly in Nampula province, noted that they had seen a decrease in the amount of learning material in Portuguese since the introduction of bilingual education.

Apart from textbooks, shortages and poor condition of classrooms and desks emerged as a significant issue; these applied to both bilingual and monolingual education. Parents talked about the challenges of students sitting on the floor and too large class sizes. The absence of libraries was also brought up, which hindered the students' ability to complete homework or self-study. Many deplored that students were not allowed to take textbooks home, preventing them from studying outside school hours.

A good number of parents across focus groups expressed satisfaction with teachers delivering bilingual education and reported no notable differences in teaching quality compared with monolingual education. In focus groups that were supportive of bilingual education, parents gave credit to teachers by acknowledging their skills that were evident from their children's progress; what they had brought home from school; and from what the children had been able to teach to other family members. In

some schools, parents praised teachers who had shown patience and dedication in helping the children understand the topics in both languages.

I can see that they're well prepared, because it's noticeable from my child's results. I can see that he's studying, because nowadays you scarcely ever find a child in Grade 2 who is able to read. But he can already read in Changana. This shows that the teacher teaches well. He teaches Grade 2, but he's good. (Woman, 27 years)

However, some parents expressed a concern that teachers were not formally trained to teach in the local language. This concern was most evident in the focus groups where parents held a negative view of bilingual education. In fact, they were worried about teacher competence in both the local language and Portuguese. Teacher training was especially important for teachers who were not fully proficient in the local language used in the school. Many parents also argued that there was a need to increase the total number of bilingual teachers. Parents suspected that, for certain cohorts, the formation of only monolingual classes had been due to a lack of teachers able to teach in the local language.

Maybe there is a teacher who is prepared to teach in Changana. The others are not qualified to teach in Changana, that's why they go to monolingual. So, for this to be successful there must be teachers first. (Woman, 38 years)

Finally, there were also parents who felt that they were not in a position to assess teacher qualifications as they had not observed any teaching. Neither were they aware whether teachers had received training in bilingual pedagogy—in itself an indication of limited parental participation in school life. Many participants also said that it was the school director who evaluated teacher performance, not parents, and that parents had to trust that teachers were adequately trained.

5.6 Sociocultural aspects

Focus group discussions also explored integrative (sociocultural) motivations. They revealed that parents' attitudes towards bilingual education extended beyond learning outcomes or future employability of their children—these were often intertwined with broader sociocultural perceptions of the value and role of local languages in society. Several participants considered it crucial for their children to learn their mother tongue as a means of intergenerational communication and fostering cultural pride and preserving their language and culture.

It is of great importance that a person knows his essence. Knowing about others, but not forgetting your culture. It is very important for a person to be able to speak his language and learn through it. (Man, 45 years)

For some parents, bilingual education signalled the government's acknowledgement of the value of Mozambican languages and culture. They wanted the local language to thrive and saw bilingual

¹⁷ In our previous study, both school directors and teachers themselves felt that teachers in bilingual education were not adequately prepared and badly needed more training (Chimbutane and Reinikka 2023).

education as an important way to 'linguistic liberation'. Others stressed the importance of language diversity, even if they recognized that local languages were overlooked in the cities. Some suggested that bilingual education could bring prosperity by valuing the country's origins; they advocated for more government effort to improve the status of local languages.

Ronga is a language like Portuguese, both transmit the same information. What we ask is that this does not just end here at school because I see that Ronga can do many more things. (Man, 37 years)

Proponents of bilingual education often emphasized the importance of children learning the local language properly, ensuring the ability to communicate with those who only speak the local language, such as elders and relatives. In some schools, participants were pleased that their children were learning the 'pure version' of a local language (also Chimbutane 2011) and could translate or teach new vocabulary to the rest of the family. Indeed, much of the discussion on potential benefits focused on the out-of-school advantages of knowing both languages.

Others, especially those who felt marginalized because of the inferior status of their language, were often less supportive of bilingual education. In multiple schools, parents talked about situations where they had felt shame or inferiority because of their inability to speak Portuguese. The prospect that their child would not learn Portuguese as quickly and as well as other children was a concern. Moreover, some parents worried that bilingual education, by prioritizing local languages over Portuguese proficiency, reinforces rural communities' sense of inferiority. Unsurprisingly, these parents supported monolingual education.

These observations highlight the interplay of integrative and instrumental motivations in defining attitudes towards the use of local languages in education. In another study, focusing on urban middle-class bilingual parents, Chimbutane and Gonçalves (2023) also show that family language decisions were driven by a combination of integrative and instrumental motivations. However, instrumental factors—socioeconomic and political power—had a higher weight in these decisions, as parents chose to invest in the transmission of Portuguese to their children at the expense of Bantu languages.

6 Discussion

Evidence from bilingual programmes based on low-status languages—such as Mozambican languages—shows that integrative motivations (sociocultural gains) tend to promote positive parental attitudes. Instrumental motivations (socioeconomic mobility), instead, tend to generate negative parental attitudes towards these programmes. Parents prefer L2-based monolingual programmes, thought to harness the cultural capital necessary for improving their children's socioeconomic status. As highlighted in Section 2.2, instrumental motivations are often found to dominate parents' attitudes in these contexts.

Interestingly, in our focus groups, positive parental attitudes towards bilingual education were more common than negative. Yet, there was considerable variation between schools, pointing to factors that can help mitigate or reverse negative attitudes. What mitigating factors did the focus groups reveal? First, there is an important role for the school in ensuring that parents receive full and regular information on bilingual education. By comparing the schools with more accurate information to those

with less (or even misinformation), it became clear that information with community engagement can help shape parents' attitudes and enhance acceptance of bilingual education.

Many parents felt that schools had not made enough effort to provide sufficient information on the rationale, processes, and benefits of bilingual education. These perceptions suggest that one-off campaigns held when the bilingual programme was first introduced is not enough to ensure buy-in. Such campaigns require systematic follow-up and reinforcement at the school level. Lack of information had led some parents to assume that only local language was used throughout primary school. Some even doubted the intentions of the programme, suspecting that their children were denied access to Portuguese language and, therefore, to meaningful education, socioeconomic mobility, and power. As discussed in Section 2, similar sentiments have also been observed in other countries in sub-Saharan Africa and elsewhere.

Given these findings, providing regular and evidence-based information on bilingual education could be an effective strategy to get parents' and entire communities' buy-in and support. The school has a key role here, with support from the district, province, and national government. With a better understanding of the bilingual modality and its pedagogical benefits, parental attitudes could become less driven by sociocultural attitudes toward local language and its current role in society.

Second, parents' attitudes were considerably more positive towards bilingual education when they had seen that their children were learning to read in the local language in early grades and, especially, when they had learnt to speak and read Portuguese in upper grades. Indeed, there was a consensus among parents that adequate command of Portuguese was crucial for their children's job market prospects. Without exception, parents who expressed reservations or were outright opposed to bilingual education insisted on the importance of Portuguese. Many argued that students in bilingual education did not attain an adequate level of Portuguese required for a job. Even supporters of the bilingual programme expressed these concerns.

This finding is consistent with the construct of Portuguese as a gateway to educational success and socioeconomic mobility as well as with the view of the school as the hub for the distribution (or denial) of this important resource (see also Chimbutane 2011, 2013; Mataruca 2014). The fact that some parents had changed their minds about bilingual education when they had witnessed progress in their children's learning also demonstrates that parental attitudes are not static.

Stronger parental support requires that learning results improve and become more evident to them—including their children acquiring proficiency in Portuguese—so that parents would trust bilingual education to deliver the results they expect. The desire to learn (in) local languages can be enhanced if communities can see that knowledge and skills acquired through these languages can amplify employment opportunities and economic dividends. As Chimbutane (2011) has shown, the use of local languages in the formal context of schools is already influencing parents' re-evaluation of the socioeconomic value of these languages. However, much more needs to be done in policy and practice to institutionalize and expand this opening. It will not happen without stronger learning results—in both L1 and L2—that the parents can themselves observe.

Parents' vision of Portuguese as the ultimate target in education can be linked to colonial and post-colonial language ideologies that have positioned Portuguese as a high-status language and local languages as inferior. Given this uneven status of languages, some parents perceive that bilingual education rather restricts children from education and puts them into a weaker position relative to

monolingual students (see also Chimbutane 2013; Chimbutane et al. 2022). Our findings indicate that these concerns often arise in instances where parents are under the impression that the sole purpose of bilingual education is to learn (in) a local language and that of monolingual education is to learn (in) Portuguese. The evidence we have produced suggests action from education and political perspectives.

How well is the Mozambican education system prepared, especially its schools and teachers, to step up delivery of quality bilingual education? Teacher training is in Portuguese, and all teachers are expected to be fully conversant in it. This is not always the case, however, as demonstrated by nationally representative school surveys, which in 2014 and 2018 assessed primary school teachers—teaching Grades 3 and 4—for their knowledge in Portuguese (Bassi et al. 2019; Molina and Martin 2015). The surveys found that teachers mastered only 32 per cent (2014) and 41 per cent (2018) of what is required to teach the lower primary curriculum in Portuguese—a cause for concern from the viewpoint of what parents expect for their children in terms of learning Portuguese either in bilingual or monolingual (Portuguese-only) classes. There was no significant skills difference between rural and urban teachers or between regions.

Teacher preparedness is even more critical for bilingual education as most of the teachers in the programme have received no pre-service training in bilingual instruction—only perhaps a short inservice training course—thus lacking the required competencies in the local language and bilingual pedagogies.

As parents have little to do with their children's classroom activities, they had relatively little to say about bilingual teaching or learning and teaching materials. Yet, they expected learning results, which require teacher preparedness and access to quality materials. While frontline actors on the supply side of education—teachers, schools directors, and local education officers—repeatedly raised the resistance of parents—the demand side—as a serious constraint for bilingual education (Chimbutane and Reinikka 2023), a joint analysis of supply and demand suggests that constraints are likely to arise primarily from weak implementation (supply). Teachers face a lack of materials and proper training in their attempt to deliver bilingual education; parents judge it based on results they themselves manage to observe; while school directors are caught in-between and depend on the rest of the education system. Although parental support is a critical element, the joint analysis underscores the fact that successful bilingual education is not possible without effective implementation to produce real learning results which, in turn, shape parental buy-in.

In brief, for a successful scale-up and sustainability of bilingual education, both effective implementation (supply side) and parental support and buy-in (demand side) are essential. Our earlier study examined the supply side—the roles, attitudes, and actions of Mozambican school directors, teachers, and local education officials—whereas this paper explored parents' attitudes on the demand side. We found that supply and demand are interdependent and both are required for a successful bilingual programme. Based on the findings of this study, we highlight two actions on the demand side that can make a positive difference: the school (i) providing full and regular information to parents and the surrounding community, and (ii) ensuring that learning results be made as visible to parents as possible. The rest of the education system should support the schools in these undertakings.

7 Summary and conclusions

Foundational learning—basic skills of reading, writing, and numeracy—is exceedingly weak in Mozambique as evident from the national learning assessments. While many factors affect learning, this paper explored bilingual education—using home language for instruction in early grades—which can offer one remedy. Specifically, evaluations of experimental bilingual programmes in several sub-Saharan African countries, including Mozambique, have shown large positive impacts on learning (Section 2.1). Moreover, an analysis of nationally representative school survey data showed that Portuguese-speaking students benefited significantly from teachers' content knowledge, but the average Mozambican student, speaking only a local language when entering school, did not. Thus, research evidence confirms that the language of instruction matters for learning.

However, the experiments are typically externally funded, short-term, and implemented by non-governmental actors. A key question, therefore, is: How to mainstream bilingual education and, while doing so, how to ensure that it delivers foundational learning at scale? One important ingredient in successful mainstreaming is parental buy-in and community mobilization—the topic of this study.

In focus group discussions in primary schools in Maputo and Nampula provinces, we first asked parents (directly) about their attitudes towards bilingual education. We then explored factors that can shape these attitudes: information; expectations about their children's school performance and labour market prospects; implementation of the bilingual programme; and general attitudes (sociocultural aspects) towards local language.

Parents' attitudes toward bilingual education varied across schools. About half of the focus groups expressed an unambiguously positive attitude towards bilingual education. A few others, while also positive, expressed caveats. Many supportive participants reported that their initially critical attitude had shifted. As experience of bilingual education accumulated, they had come to accept it. Yet, there were a few schools where parents were sceptical and outright resisted bilingual education, particularly in Nampula province.

In terms of factors shaping parental attitudes, focus group discussions revealed that, due to differing practices in schools, the *information* parents possessed on bilingual education was highly variable, inconsistent, and often incomplete. In some schools, parents understood and appreciated its pedagogical rationale, that is, that the use of local language makes learning of foundational skills easier. In other schools misconceptions prevailed, creating or reinforcing scepticism and resistance. In schools where parents had received more information (relative to other schools)—as can be expected—parents were more supportive of bilingual education. Participants insisted that the school needs to provide information to the whole community, not just parents whose children are in bilingual classes. All in all, our findings highlight that information is a critical element in shaping parents' attitudes and acceptance of bilingual education.

Perhaps the most important influence on parents' attitudes, across the schools, was how well their children were *learning*—a key finding also in earlier studies in Mozambique (e.g., Benson 2000). Especially, when parents had themselves observed that their children were able to read in the local language in early grades, they were supportive of bilingual education. The most supportive were those parents whose children had learnt to speak and read Portuguese in upper grades.

In schools where parents expressed reservations or where opposition to bilingual education was strong, parents insisted that they had enrolled their children in school for them to learn Portuguese. Negative experiences among parents, such as witnessing a child going through bilingual education without learning properly either the local language or Portuguese, increased parental pressure on the school to switch back to monolingual education. A good number of parents reported that they had not seen sufficient Portuguese acquisition in bilingual classes, placing their children in an unfavourable position relative to children coming from monolingual education.

Concerns about sufficient Portuguese acquisition made parents doubtful about the overall benefits of bilingual education even in cases where they accepted its pedagogical rationale. This was largely driven by socioeconomic aspirations. Mastering the dominant language, Portuguese, was considered a must for those aspirations to materialize, especially in the job market. A common refrain among parents was: 'Portuguese is needed for getting a job.' No other subject or foundational skill—say, numeracy—came up as a pre-requisite for finding a job.

Parents' perceptions on implementation of the bilingual programme were quite different from those of school directors, teachers, and local education officials, which we explored in our earlier study in Maputo province. Teachers had been especially critical about scarcity and poor quality of teaching and learning materials, as well as absence of training opportunities in bilingual pedagogy. Parents did not mention teaching resources at all, which is not surprising as they had little or no direct engagement with teaching or classroom activities. Regarding learning materials, a majority of parents did not see much difference between bilingual and monolingual (Portuguese-only) education. However, a general lack of textbooks was a common concern to focus group participants, especially Portuguese textbooks in upper grades. Equally, across the schools, parents complained about shortages of classrooms and school furniture.

While recognizing a lack of training opportunities for teachers, overall, parents gave a lot of credit to their children's teachers, including those teaching bilingual classes. Indeed, many interventions in focus group discussions revealed respect for teachers.

Our earlier study showed that it is school directors who decide whether or not to establish bilingual classes and how to allocate the incoming students to different classes. School directors argued that there was strong pressure from parents that affected these decisions. Contrary to this argument, few parents participating in the focus groups mentioned such pressure—and when they had been asked about local language by the school, it had not had impact on their children's placement.

While integrative (sociocultural) motivations were important to many participants, they were not enough. Academic benefits and learning of Portuguese, which were expected to lead to instrumental (socioeconomic) gains, were parents' top priority, including parents who were supportive of bilingual education. For some the expansion of bilingual education signalled the government's acknowledgement of the value of Mozambican languages and culture. They wanted local languages to thrive and saw bilingual education as an important way of 'linguistic liberation'. Others, especially those who felt marginalized because of the inferior status of their language, were less supportive of bilingual education.

In conclusion, this study underscores the importance of learning results that parents can themselves observe. To ensure broader acceptance and support, it is crucial for schools consistently to communicate the benefits of bilingual education as well as address parental concerns, especially

Portuguese proficiency. Bridging the gap between the educational advantages of bilingual programmes and parents' socioeconomic goals for their children is key to the success and sustainability of bilingual education in Mozambique.

References

- Ball, M.-C., J. Bhattacharya, H. Zhao, H. Akpé, S. Brogno, and K. Jasińska (2024). 'Effective Bilingual Education in Francophone West Africa: Constraints and Possibilities'. *International Journal of Bilingual Education and Bilingualism*, 27(6): 821–35. https://doi.org/10.1080/13670050.2023.2290482
- Bamgbose, A. (1999). 'African Language Development and Language Planning'. Social Dynamics, 25(1): 13–30. https://doi.org/10.1080/02533959908458659
- Banda, F. (2000). 'The Dilemma of Mother Tongue: Prospects for Bilingual Education in South Africa'. *Language, Culture and Curriculum*, 13(1): 51–66. https://doi.org/10.1080/07908310008666589
- Bassi, M., O. Medina Pedreira, and L.C. Nhampossa (2019). 'Education Service Delivery in Mozambique: A Second Round of the Service Delivery Indicators Survey'. Report. Washington, DC: World Bank Group. Available at: https://documents.worldbank.org/en/publication/documents-reports/documentdetail/811891562864504006/education-service-delivery-in-mozambique-a-second-round-of-the-service-delivery-indicators-survey (accessed 9 December 2024).
- Benson, C.J. (2000). 'The Primary Bilingual Education Experiment in Mozambique, 1993 to 1997'. *International Journal of Bilingual Education and Bilingualism*, 3(3): 149–66. https://doi.org/10.1080/13670050008667704
- Bunyi, G. (2008). 'Constructing Elites in Kenya: Implications for Classroom Language Practices in Africa'. In N.H. Hornberger (eds), *Encyclopedia of Language and Education*. Boston, MA: Springer, pp. 899–909. https://doi.org/10.1007/978-0-387-30424-3_68
- Chiatoh, B.A. (2011). 'Sustaining Mother Tongue Medium Education: An Inter-Community Self-Help Framework in Cameroon'. *International Review of Education*, *57*: 583–97. https://doi.org/10.1007/s11159-011-9247-3
- Chimbutane, F. (2011). Rethinking Bilingual Education in Postcolonial Contexts. Bristol, UK: Multilingual Matters. https://doi.org/10.21832/9781847693655
- Chimbutane, F. (2013). 'Can Sociocultural Gains Sustain Bilingual Education Programs in Postcolonial Contexts? The Case of Mozambique'. In J.A. Shoba and F. Chimbutane (eds), *Bilingual Education and Language Policy in the Global South.* London: Routledge, pp. 124–45.
- Chimbutane, F. (2015). 'Bilingual Education: Enabling Classroom Interaction and Bridging the gap Between Schools and Rural Communities in Mozambique'. *International Journal of Educational Development in Africa*, 2(1): 101–120. https://doi.org/10.25159/2312-3540/19
- Chimbutane, F. (2021). Bilingual Education and Multilingualism in Mozambique: A Decolonial Critique of Policies, Discourses and Practices. In K. Heugh, C. Stroud, K. Taylor-Leech, and P. De Costa (eds), *A Sociolinguistics of the South,* pp. 138–154. London & New York, NY: Routledge. https://doi.org/10.4324/9781315208916-12
- Chimbutane, F.; J. Ennser-Kananen, and S. Kosunen (2022). 'The socio-material value of language choices in Mozambique and Finland'. In J. Ennser-Kananen and T. Saarinen (eds), *New materialist explorations into language education*, pp. 111–132. Gewerbestrasse, Switzerland: Springer. (eBook). https://doi.org/10.1007/978-3-031-13847-8
- Chimbutane, F., and Gonçalves, P. (2023). 'Family Language Planning and Language Shift in Postcolonial Mozambique'. *Language Policy*, 22: 267–88. https://doi.org/10.1007/s10993-023-09658-3
- Chimbutane, F., and R. Reinikka (2023). "Language and Student Learning: Evidence from an Ethnographic Study in Mozambique." WIDER Working Paper 2023/62 Helsinki: UNU-WIDER, 2023. https://doi.org/10.35188/UNU-WIDER/2023/370-3
- Chonane, D.Z. (2024). Percepções e atitudes dos pais e encarregados de educação sobre a educação bilingue no distrito da Manhiça [Perceptions and Attitudes of Parents and Guardians Towards Bilingual Education in Maniça District]. Unpublished PhD Thesis, Universidade Eduardo Mondlane.
- Cumbane, S. S. (2020). 'Educação bilingue em Moçambique e o empoderamento comunitário' [Bilingual Education in Mozambique and Community Empowerment]. *Multilingual Margins*, 7(1): 11–29. https://doi.org/10.14426/mm.v7i1.1376

- Gupta, A.F. (1997). 'When Mother-Tongue Education Is Not Preferred. *Journal of Multilingual and Multicultural Development*, 18(6): 496–506. https://doi.org/10.1080/01434639708666337
- Hanushek, E.A., M. Piopiunik, and S. Wiederhold (2019). 'The Value of Smarter Teachers: International Evidence on Teacher Cognitive Skills and Student Performance'. *Journal of Human Resources*, 54(4): 857–99. https://doi.org/10.3368/jhr.54.4.0317.8619R1
- Henriksen, S.M. (2014). 'Ideologies of Language and Bilingual Education in Mozambique'. In K. Juffermans, Y.M. Asfaha, and A. Abdelhay (eds), *African Literacies: Ideologies, Scripts, Education*. Newcastle: Cambridge Scholar Publishing.
- Heugh, K. (2009). 'Into the Cauldron: An Interplay of Indigenous and Globalised Knowledge with Strong and Weak Notions of Literacy and Language Education in Ethiopia and South Africa'. *Language Matters*, 40(2): 166–89.
- Heugh, K., C. Benson, B. Bogale, and M.A.G. Yohannes (2007). 'Final Report. Study on Medium of Instruction in Primary Schools in Ethiopia'. Commissioned by the Ministry of Education. Addis Ababa: Ministry of Education of Ethiopia.
- Holvio, A. (2022). 'Impact of Teacher Content Knowledge on Student Achievement in a Low-income Country'. WIDER Working Paper 2022/23. Helsinki: UNU-WIDER. https://doi.org/10.35188/UNU-WIDER/2022/154-9
- Hovens, M. (2002). 'Bilingual Education in West Africa: Does It Work?'. *International Journal of Bilingual Education and Bilingualism*, 5(5): 249–66. https://doi.org/10.1080/13670050208667760
- INDE/MINEDH (2014). Avaliação Nacional da 3ª Classe. Maputo: Instituto Nacional do Desenvolvimento da Educação/Ministério da Educação e Desenvolvimento Humano (INDE/MINEDH).
- INDE/MINEDH (2017). Avaliação Nacional da 3ª Classe. Maputo: Instituto Nacional do Desenvolvimento da Educação/Ministério da Educação e Desenvolvimento Humano (INDE/MINEDH).
- Instituto Nacional de Estatística. (2019). IV Recenseamento geral da população e habitação 2017: Resultados definitivos [IV General Population and Housing Census 2017: Final Results]. Instituto Nacional de Estatística.
- Kitoko-Nsiku, E. (2007). 'Dogs' Languages or People's Languages? The Return of Bantu Languages to Primary Schools in Mozambique'. *Current Issues in Language Planning*, 8(2): 258–82. https://doi.org/10.2167/cilp111.0
- Krueger, R.A., and A.M. Casey (2009). Focus Groups: A Practical Guide for Applied Research, 4th edition. Thousand Oaks, CA: SAGE.
- Laitin, D.D., R. Ramachandran, and S.L. Walter (2019). 'The Legacy of Colonial Language Policies and Their Impact of Student Learning: Evidence from an Experimental Program in Cameroon'. *Economic Development and Cultural Change*, 68(1): 239–72. https://doi.org/10.1086/700617
- Manuel, C.J., F. Chimbutane, C. Lauchande, G. Chambo, T.J. D'Agostino, C. Freeman, and A. Conaghan, (2024). 'Language of Instruction Translation in Education Systems'. LITES Mozambique Final Report. Maputo: Supporting Holistic and Actionable Research in Education (SHARE): USAID.
- Manyonyi J., B. Mbori, and E. Okwako (2016) 'Attitude of Teachers Towards Use of Mother Tongue as Medium of Instruction in Lower Primary Schools in Bungoma South Sub-County, Kenya'. *International Journal of Education and Research*, 4(8). https://www.ijern.com/journal/2016/August-2016/24.pdf
- Mataruca, C.Q. (2014). Raising Literacy Levels in Mozambique: The Challenges of Bilingual Education in a Multilingual Post-Colonial Society. PhD Thesis, University of Southern Queensland, Australia.
- MINEDH (2019). *Estratégia de Expansão do Ensino Bilingue (EEEB) 2020–2029* [Bilingual Education Expansion Strategy (EEEB) 2020–2029]. Maputo: Ministério da Educação e Desenvolvimento Humano (MINEDH).
- MINEDH (2020). Plano Curricular do Ensino Primaria: Objetivos, Política, Estrutura, Plano de Estudos e Estratégias de Implementação [Primary Education Curriculum Plan: Objectives, Policy, Structure, Study Plan and Implementation Strategies]. Instituto Nacional de Desenvolvimento da Educação. Maputo: Ministério da Educação e Desenvolvimento Humano (MINEDH).
- MINEDH (2022). Estatísticas da Educação Levantamento Escolar [Education Statistics: School Survey]. Maputo: Ministério da Educação e Desenvolvimento Humano (MINEDH).
- Mohohlwane, N., S. Taylor, J. Cilliers, and B. Fleisch (2024). 'Reading Skills Transfer Best from Home Language to a Second Language: Policy Lessons from Two Field Experiments in South Africa'. *Journal of Research on Educational Effectiveness*, 17(4): 687–710. https://doi.org/10.1080/19345747.2023.2279123
- Molina, E., and G. Martin (2015). 'Education Service Delivery in Mozambique'. Service Delivery Indicators (SDI). Washington, DC: World Bank Group.

- Moses, M.S. (2000). 'Why Bilingual Education Policy Is Needed: A Philosophical Response to the Critics'. Bilingual Research Journal, 24(4): 333–54. https://doi.org/10.1080/15235882.2000.10162771
- Ngunga, A. (2011). 'Monolingual Education in a Multilingual Setting: The Case of Mozambique'. *Journal of Multicultural Discourses*, 6(2): 177–96. https://doi.org/10.1080/17447143.2011.577537
- Ngunga, A., N. Nhongo, L. Moisés, J. Langa, H. Cherinze, and J. Mucavele (2010). *Educação Bilingue na Província de Gaza: Avaliação de um Modelo de Ensino* [Bilingual Education in Gaza Province: Evaluation of a Teaching Model]. Maputo: Centro de Estudos Africanos, Universidade Eduardo Mondlane.
- Ouane, A., and C. Glanz (2010). Why and How Africa Should Invest in African Languages and Multilingual Education: An Evidence- and Practice-Based Policy Advocacy Brief. UNESCO Institute for Lifelong Learning. Available at: https://unesdoc.unesco.org/ark:/48223/pf0000188642 (accessed April 2025).
- Ouane A., and C. Glanz (eds). (2011). Optimizing Learning, Education and Publishing in Africa: The Language Factor. A Review and Analysis of Theory and Practice in Mother-Tongue and Bilingual Education in Sub-Saharan Africa. UNESCO Institute for Life Long Learning and Association for the Development of Education in Africa (ADEA). Available at: https://files.eric.ed.gov/fulltext/ED540491.pdf (accessed April 2025).
- Parker, A., and J. Tritter (2006). 'Focus Group Method and Methodology: Current Practice and Recent Debate'. International Journal of Research & Method in Education, 29(1): 23–37. https://doi.org/10.1080/01406720500537304
- Patel, S.A. (2006). 'Educação bilingue em Moçambique: Um olhar sobre o professor e sua prática docente' [Bilingual Education in Mozambique: A Look at the Teacher and the Teaching Practice]. *Aquí Estamos*, México, Ano 2, 4: 70–84. Available at: http://www.ciesas.edu.mx/ciesas-ford/Revistas.html (accessed April 2025).
- Phindane, P. (2015). 'Learning in Mother Tongue: Language Preferences in South Africa'. *International Journal of Educational Sciences*, 11(1): 106–11 http://dx.doi.org/10.1080/09751122.2015.11890380
- RISE Research and Communications (2023). 'Foundational Skills'. RISE Programme. Oxford: Blavatnik School of Government, University of Oxford. Available at: https://riseprogramme.org/systems-thinking/foundational-skills.html (accessed April 2025).
- Rubagumya, C.M. (2003). 'English Medium Primary Schools in Tanzania: A New "Linguistic Market" in Education'. In B. Brock-Utne, Z. Desai, M. Qorro, and A. Pitman (eds), *The Language of Instruction in Tanzania and South Africa* (LOITASA), Dar es Salaam: E&D Ltd. https://doi.org/10.1163/9789460912221_005
- Sridhar, K.K. (1994). 'Mother Tongue Maintenance and Multiculturalism'. TESOL Quarterly, 28(3): 628–31. https://doi.org/10.2307/3587315
- Stroud, C. (2001). 'African Mother-Tongue Programmes and the Politics of Language: Linguistic Citizenship Versus Linguistic Human Rights'. *Journal of Multilingual and Multicultural Development*, 22(4): 339–55. https://doi.org/10.1080/01434630108666440
- Stroud, C. (2002). Towards a Policy for Bilingual Education in Developing Countries. Stockholm: SIDA, Education Division.
- Sumich, J. (2018). The Middle Class in Mozambique. The State and the Politics of Transformation in Southern Africa. Cambridge: International African Library, Cambridge University Press. https://doi.org/10.1017/9781108659659
- Terra, S.E.L. (2021). 'Bilingual Education in Mozambique: A Case-Study on Educational Policy, Teacher Beliefs, and Implemented Practices'. *International Journal of Bilingual Education and Bilingualism*, 24(1): 16-30. https://doi.org/10.1080/13670050.2018.1441803
- Trudell, B., and B. Piper (2014). 'Whatever the Law Says: Language Policy Implementation and Early-Grade Literacy Achievement in Kenya'. *Current Issues in Language Planning*, 15(1): 4–21. https://doi.org/10.1080/14664208.2013.856985
- UNESCO Institute for Statistics. 2017. 'Counting the Number of Children Not Learning. Methodology for a Global Composite Indicator for Education'. *Information Paper* No. 47.
- UNESCO (2023). Spotlight on Basic Education Completion and Foundational Learning in Mozambique. Paris.
- USAID (2017). Language Mapping Study in Mozambique. Maputo.
- USAID (2022). Vamos Ler! LEMA 2021 Report. Maputo.
- Van Ingelgom, V. (2020). 'Focus Groups: From Qualitative Data Generation to Analysis'. In L. Curini and R. Franzese (eds), *The SAGE Handbook of Research Methods in Political Science and International Relations*. London: SAGE, pp. 1190–210. https://doi.org/10.4135/9781526486387.n65

- Walter, S.L., and C. Benson (2012). 'Language Policy and Medium of Instruction in Formal Education'. In B. Spolsky (ed.), *The Cambridge Handbook of Language Policy* (1st ed.). Cambridge: Cambridge University Press, pp. 278–300. https://doi.org/10.1017/CBO9780511979026.017
- World Bank (n.d.). Service Delivery Indicators. Available at: https://www.worldbank.org/en/programs/service-delivery-indicators (accessed April 2025).
- World Bank (2017). World Development Report 2018: Learning to Realize Education's Promise. Washington, DC: World Bank.